Author's response to reviews

Title: WEB downloadable software for training in cardiovascular hemodynamics in the (3-D) stress echo lab

Authors:

Tonino Bombardini (bombardini@ifc.cnr.it)
Davide Cini (davide.cini@cnr.it)
Giorgio Arpesella (arpesella@aosp.bo.it)
Eugenio Picano (picano@ifc.cnr.it)

Version: 3 Date: 7 November 2010

Author's response to reviews: see over
Editor-in-Chief
Eugenio Picano (Italy)

Deputy Editor
Rosa Sicari (Italy)

Cardiovascular ultrasound

Pisa, November 7, 2010

Author list: Tonino Bombardini, Davide Cini, Giorgio Arpesella and Eugenio Picano
Title: WEB downloadable software for training in cardiovascular hemodynamics in the (3-D) stress echo lab
Journal: Cardiovascular Ultrasound
MS ID: 1832638701459128

We made the following changes to the format of our manuscript, so that the published article conforms to journal style.

Major revisions

References:
References are cited in numerical order.
Reference 51 is now cited within the main text.
All references in the text are cited by the corresponding number in the reference list.

The new Figure 6 is now smaller according to your suggestion.

Minor revisions

We removed from the abstract in the manuscript: Keywords.

We formatted all red text to black.

We formatted the abbreviations in to a paragraph.

We moved the disclaimer to the Acknowledgements section.

We made sure that the table title is placed above the table and not in the table.
We placed the figure legends and the tables after the reference list.

Looking forward to hearing from you.

Sincerely

Tonino Bombardini, MD